\begin{table}%t3
\caption{\label{t:doubles}Multiple entries in the Brahe catalogue.}
\par
%\centering
\par
\begin{tabular}{l@{\hspace{0.1cm}}rrrrcl}
\hline\hline \noalign{\smallskip}
& \phantom{KK}K & K(M) & HIP\phantom{I} & $d_{1,2}$(\arcmin) & $\Delta M$ & See\\
\hline
 &  10 & 252 & 23734 &   0.0 & 0 \\
a &  14 &  64 & 51808 &  13.8 & 0 & Fig.~\ref{f:ursaminor} near +6, +2  \\
 & 201 & 216 & 96441 &   7.8 & 0 & Fig.~\ref{f:cygdetail} \\
 &     & 220 & 96441 &   1.0 & 2 & Fig.~\ref{f:cygdetail} \\
 & 333 & 360 & 80883 &   6.6 & 1 & Fig.~\ref{f:ophiuchus} near $-$6, +6 \\
 & 336 & 347 & 86284 &  46.3 & 0 & Fig.~\ref{f:ophdetailb} \\
 &     & 364 & 86284 &  30.0 & 1 & Fig.~\ref{f:ophdetailb} \\
 & 339 & M701 & 84012 &  0.0 & 0 \\
a & 341 & 353 & 84893 &  22.3 & 0 & Fig.~\ref{f:ophdetail} near +8, $-$16 \\
 & 344 & 361 & 84880 &  11.2 & 1 & Fig.~\ref{f:ophdetailb} \\
 & 345 & 362 & 86263 &   5.1 & 1 & Fig.~\ref{f:ophdetailb} \\
 & 346 & 363 & 86565 &  16.5 & 1 & Fig.~\ref{f:ophdetailb} \\
 & 470 & 483 &  --   &   0.0 & 0 \\
 & 584 & M94 & 39780 &   3.6 & 1 \\
 & 908 & M913 & 17593 &  0.4 & 0 \\
b & 13 & 297 & 15520 & -- & 0 \\
b & 738 & 739 & 108036 & -- & 0 \\
b & \multicolumn{2}{l}{205-7\phantom{777}219} & 102589 & -- & 2 \\
\hline
\end{tabular}
\tablefoot {For each pair of K (or M) numbers the table lists the {\it Hipparcos} identification, with the difference in positions and magnitude of the Brahe entries and the approximate position in a figure in Sect.~\ref{s:figures}. Effects of position corrections by Rawlins (\cite{rawlins}, esp.~footnote~77): a: no~longer repeated entry, b: new repeated entry.}
\end{table}